Author's response to reviews

Title: Early Life Environment and Natural History of Inflammatory Bowel Diseases

Authors:

Abra Guo (aguo1@mgh.harvard.edu)
Betsy Stevens (bstevens1@partners.org)
Robin Wilson (rgwilson@mgh.harvard.edu)
Caitlin Russell (cnrussell@mgh.harvard.edu)
Melissa Cohen (mcohen4@mgh.harvard.edu)
Holly Sturgeon (hsturgeon@mgh.harvard.edu)
Anna Thornton (athornton4@mgh.harvard.edu)
Cosmas Giallourakis (cgiallourakis@mgh.harvard.edu)
Hamed Khalili (hkhalili@mgh.harvard.edu)
Deanna Nguyen (deanna.nguyen@mgh.harvard.edu)
Jenny Sauk (jsauk@partners.org)
Vijay Yajnik (vyajnik@partners.org)
Ashwin Ananthakrishnan (aananthakrishnan@partners.org)

Version: 3 Date: 18 September 2014

Author's response to reviews:

**Please move Authors Contribution, Competing Interest, Acknowledgment and Funding after Conclusions.

This has been done now.

**Request for ethics statement.

Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

This has been provided.

**Please clarify the access to the Prospective Registry in IBD Study at Massachusetts General Hospital (PRISM).

Not sure what this refers to. Access for whom?